Author’s response to reviews

Title: Profile and Factors Associated with Glycaemic Control of Patients with Type 2 Diabetes in Greece: Results from the Diabetes Registry

Authors:
KYRIAKOS SOULIOTIS (info@ksouliotis.gr;ksouliotis@uop.gr)
Anastasios Koutsovasilis (anastasioskou@gmail.com)
Georgia Vatheia (gina_vathia@yahoo.com)
Christina Golna (c.golna@innowth.com)
Sofia Nikolaidi (sofianikolaidi@icloud.com)
Erifili Hatzigelaki (erihat@med.uoa.gr)
Kalliopi Kotsa (kalmanthou@yahoo.gr)
Theocharis Koufakis (thkoyfak@hotmail.com)
Andreas Melidonis (melidonisa@yahoo.com)
Athanasia Papazafiropoulou (pathan@ath.forthnet.gr)
Nikolaos Tentolouris (ntentol@med.uoa.gr)
Evangelia Siami (evasiami@gmail.com)
Alexios Sotiropoulos (alexiosotiropoulos@gmail.com)

Version: 2 Date: 20 Jan 2020

Author’s response to reviews:

Response to Editor

Dear Editor,

Thank you very much for your letter informing us that our article is potentially acceptable for publication in BMC Endocrine Disorders. We are very grateful to you for your suggestions on our manuscript "BMC Endocrine Disorders Profile and Factors Associated with Glycaemic Control of Patients with T2DM in Greece: Results from the Diabetes Registry” [BEND-D-19-00159]. Our revisions have addressed all comments and incorporated all recommendations you have made.

Below please find our detailed responses to your comments.
On behalf of the authors,

Kyriakos Souliotis

-“My only suggestion is to advice the authors (or the print editors) to use people first language; so not to use diabetic patients or the terms obese and instead to use patients with diabetes, and patients with obesity. The same applies to the term T2DM patients.”

Amendments were made throughout text.

2. Please note that all manuscripts must contain all the following sections under the heading 'Declarations'. The Declarations should follow the Conclusions section, and be before the References.
   • Ethics approval and consent to participate
   • Consent for publication
   • Availability of data and material
   • Competing interests
   • Funding
   • Authors’ contributions
   • Acknowledgements

The authors confirm that the above declarations have now been added before the References section.